Author's response to reviews

Title: Completion Pneumonectomy for Lung Cancer Treatment: Early and Long Term Outcomes in China

Authors:

peng zhang (zhangpeng1121@yahoo.com.cn)
chao jiang (jiangchao@hotmail.com)
wenxin he (wenxinhe@hotmail.com)
nan song (nansong@hotmail.com)
xiao zhou (xiaozhou@hotmail.com)
gening jiang (zhangpeng1121@hotmail.com)

Version: 3 Date: 9 July 2012

Author's response to reviews:

Dear editor:

The two reviewers gave us many good suggestions. We have corrected our manuscript with the help of their suggestions, such as in grammar and spelling mistakes. We hope our manuscript could reach the requirement of the Journal.

Best regards

Peng Zhang